Author's response to reviews

Title: Robotic radical nephrectomy for renal cell carcinoma: a systematic review.

Authors:

Anastasios D. Asimakopoulos MD, PhD, FEBU (tasospao2003@yahoo.com)
Roberto Miano MD (mianor@virgilio.it)
Filippo Annino MD (filippo.annino@gmail.com)
Salvatore Micali MD (salvatore.micali@unimore.it)
Enrico Spera Prof (enrico.spera@uniroma2.it)
Beniamino Iorio MD (beniaminoiorio@tiscali.it)
Giuseppe Vespasiani Prof (vespasiani@med.uniroma2.it)
Richard Gaston MD (gastonrich@wanadoo.fr)

Version: 4
Date: 3 July 2014

Author's response to reviews:

This version is very good and shows a good job done on the manuscript. Nevertheless two small errors should be addressed:

1) please specify the period of the research (ie: January 2000 to December 2013) specifying the months;
   Performed, please see lines 104-105

2) please explain the acronym R-LESS before its routine use in the discussion.
   Performed, see lines 135-136

3) Please also highlight (with 'tracked changes'/coloured/highlighted text) all changes made when revising the manuscript to make it easier for the Editors to give you a prompt final decision on your manuscript. We prefer if you can make text look like it was marked with a highlighter pen the modified part of the original manuscript.

   All the changes performed in the abstract, text, tables and references are already highlighted in red.